Author's response to reviews

Title: Stepped care treatment for depression and anxiety in primary care: A randomized controlled trial

Authors:

Wike Seekles (wm.seekles@psy.vu.nl)
Annemieke van Straten (a.van.straten@psy.vu.nl)
Aartjan Beekman (a.beekman@ggzingeest.nl)
Harm van Marwijk (hwj.vanmarwijk@vumc.nl)
Pim Cuijpers (p.cuijpers@psy.vu.nl)

Version: 2 Date: 13 April 2011

Author's response to reviews: see over
April, 2011

Dear Sir/Madam,


The following issue needed my attention and I added the Tables (1-3) in the manuscript after the reference list:

Please include the Tables (1-3) within the main manuscript document after the reference list, instead of as figure files. The figures (1-2) should remain uploaded as figure files.

The tables (1-2) are uploaded as figure files.

Hereby you receive a revised version of my manuscript.

I will be serving as the corresponding author for this manuscript. All co-authors agreed to submission of the manuscript in this form. If you have any questions concerning the study, you can contact us by phone or email.

Sincerely,

Wike Seekles, PhD student
Department of Clinical Psychology
Faculty of Psychology and Education
VU University, Amsterdam
Van der Boechorststraat 1
1081 BT Amsterdam
phone: 0031 (0)20 598 2544
email: wm.seekles@psy.vu.nl